Author's response to reviews

Title: Specific gene expression profiles and chromosomal abnormalities are associated with infant disseminated neuroblastoma

Authors:

- Cinzia Lavarino (clavarino@fsjd.org)
- Nai-Kong V Cheung (cheungn@mskcc.org)
- Idoia Garcia (igarcia@fsjd.org)
- Gema Domenech (gema.domenech@h1.org.es)
- Carmen de Torres (ccd@hsjdbcn.org)
- Miguel Alaminos (malaminos@histolii.ugr.es)
- Jose Rios (jose.rios@uab.es)
- William L Gerald (geraldw@mskcc.org)
- Brian Kushner (kishnerb@mskcc.org)
- Mike LaQuaglia (laquaglia@mskcc.org)
- Jaume Mora (jmorag@hsjdbcn.org)

Version: 7 Date: 30 January 2009

Author's response to reviews: see over
Formatting changes requested

- Weblink on page 5 cannot be accessed. Please ensure the URL is correct.

The Weblink has been corrected on page 5 has been corrected to: